Author’s response to reviews

Title: Acute Versus Chronic Methotrexate Poisoning; a Retrospective Study

Authors:

Arman Ahmadzadeh (arman319@yahoo.com)
Nasim Zamani (nasim.zamani@gmail.com)
Hossein Hassanian-Moghaddam (hassanian@sbmu.ac.ir)
Seyed Kaveh Hadeiy (kaveh.hadee@yahoo.com)
Parinaz Parhizgar (p.parhizgar@yahoo.com)

Version: 3 Date: 28 May 2019

Author’s response to reviews:

May 28, 2019

Dear Prof William Spenceley Jonesaa

BMC Pharmacology and Toxicology

Please find our response to editor’s comment in which we have revised suggested parts. Editor Comments:

1. In the section 'Ethics Approval and Consent to Participate', please state whether the informed consent obtained was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.

Reply: The written informed consent was taken on arrival time. The ethic’s committee confirmed using retrospective data based on this consent.

2. Currently, the age and gender information in Table 1, in conjunction with other details, may compromise patient anonymity. If your manuscript contains any individual person’s data in any form (including individual details, images or videos), consent to publish must be obtained from that person/next-of-kin, or in the case of children, their parent or legal guardian. Please provide details of this consent in this section, including whether the informed consent obtained was written or verbal. Please see our editorial policies for more information - https://www.biomedcentral.com/getpublished/editorial-policies#consent+for+publication
you do not have consent to publish please amend table 1 to address this issue, by providing ages as age-ranges, or removing the age (or sex) information.
Reply: We amend table 1 and include an age range to keep patient’s anonymity

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional-supplementary files are cited within the text.

Reply: Done

Yours truly,

Hossein Hassanian-Moghaddam MD, FACMT

Corresponding Author